Author’s response to reviews

Title: What He Knows About Her And How It Affects Her? Husband's Knowledge of Pregnancy Complications and Maternal Health Care Utilization Among Tribal Population in Maharashtra, India

Authors:

Suresh Jungari (sureshjungariiips@gmail.com)
Balram Pawan (bpaswaniips@gmail.com)

Version: 2 Date: 28 Jan 2019

Author’s response to reviews:

We authors thanks for the thorough review of our manuscript (PRCH-D-18-00811). We have revised our manuscript accordingly. Our point-by-point responses are provided below, and the changes to the manuscript within the document have been track change mode. We hope our efforts to response to all the comments raised by reviewers have addressed.

Reviewer reports:

Solwayo Ngwenya, MBChB, DFSRH, FRCOG (Reviewer 1): Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

Please overwrite this text when adding your comments to the authors. Nil

Authors response: Thank you for review

Heidi Preis (Reviewer 3):

Comment: I commend authors for the thorough revisions they made which have much improved the manuscript. The only issue I would like to comment on is that the categories of age and
education in Table 1 are different than those in the other tables. I suggest correcting the categories in Table 1 to correspond with the rest.

Author’s response: We have corrected age and education categories in the table 1. Thank you.